Author’s response to reviews

Title: Standardization of the Colombian Version of the PHQ-4 in the general population

Authors:

Rüya-Daniela Kocalevent (r.kocalevent@uke.de)
Carolyn Finck (cfinck@uniandes.edu.co)
William Jimenez-Leal (wa.jimenez20@uniandes.edu.co)
Leon Sautier (l.sautier@uke.de)
Andreas Hinz (andreas.hinz@medizin.uni-leipzig.de)

Version: 4 Date: 10 July 2014

Author’s response to reviews:

In this revision we included the email addresses of the authors and the acknowledgements according to the suggestion of the editor.